Author's response to reviews

Title: Comparative evidence for a link between Peyer's patch development and susceptibility to transmissible spongiform encephalopathies

Authors:

Suzanne G St.Rose (s0231156@sms.ed.ac.uk)
Nora Hunter (nora.hunter@bbsrc.ac.uk)
Louise Matthews (l.matthews@vet.gla.ac.uk)
James D Foster (jim.foster@bbsrc.ac.uk)
Margo E Chase-Topping (margo.chase@ed.ac.uk)
Loeske EB Kruuk (Loeske.Kruuk@ed.ac.uk)
Darren J Shaw (Darren.Shaw@ed.ac.uk)
Susan M Rhind (Susan.Rhind@ed.ac.uk)
Robert G Will (r.g.will@ed.ac.uk)
Mark EJ Woolhouse (Mark.Woolhouse@ed.ac.uk)

Version: 3 Date: 21 December 2005

Author's response to reviews:

Changes made to revised manuscript

1. Authors
   E-mail address and contact address for one of the co-authors (Louise Matthews) has been changed accordingly (both in the final manuscript and the submission system).

2. Figures
   The figure numbers have been removed from the respective image files.

3. Author's contributions
   Author's contributions has been changed to Authors' contributions

4. Formatting
   Indentations have been removed and individual paragraphs separated.